CS2LAP: Logic and Prolog 2000/2001 Labs and office hours

Labs

Labs sessions will take place roughly every other week in room 24D. Turn up at one of the following times:

• Tuesdays, 5-6pm • Thursdays, 2-3pm (unless this slot is used for DB, in which case we move to 3-4pm) The first lab sessions are going to be next week, i.e. on October the 3rd and the 5th, respectively.

Ulle Endriss, King’s College London 1

CS2LAP: Logic and Prolog 2000/2001 Labs and office hours

Office Hours

Antonio de Freitas Ulle Endriss Email: antonio@dcs.kcl.ac.uk Email: endriss@dcs.kcl.ac.uk Room 542 (Strand, 5th floor) Room 542 (Strand, 5th floor) Office hours: Thu 2-4pm Office hours: Thu 2-4pm (minus lab time)

If you can’t make it Thursday afternoon please send an email to make an appointment.

Ulle Endriss, King’s College London 2